Total score /100 pts. Student name Markers

SE 2A04 Fall 2001 Lab Exercise 1 Marking Scheme

Instructor: William M. Farmer

1. Demonstration of test program /20 pts. 2. Report

(a) Specification

i. Specification statement /5 pts. ii. Discussion /10 pts.

(b) Program that implements specification

i. Code /5 pts. ii. Explanation /10 pts.

(c) Test program

i. Code /5 pts. ii. Explanation /10 pts.

(d) Testing results

i. Results on testing your program /10 pts. ii. Results on testing your partner’s program /10 pts.

(e) Discussion of the exercise and test results /10 pts. (f) Log /5 pts.